Author’s response to reviews

Title: Health-related rehabilitation services: assessing the global supply of and need for human resources

Authors:

Neeru Gupta (neeru.gupta@gnb.ca)
Carla Castillo Laborde (carlacastillo@minsal.cl)
Michel D Landry (mike.landry@duke.edu)

Version: 4 Date: 14 October 2011

Author’s response to reviews:

Final edits as requested.